Author's response to reviews

Title: Three-year follow-up of physical activity in Norwegian youth from two ethnic groups: associations with socio-demographic factors

Authors:

Åse Sagatun (aase.sagatun@r-bup.no)
Elin Kolle (Elin.Kolle@nih.no)
Sigmund A. Anderssen (sigmund.anderssen@nih.no)
Magne Thoresen (magne.thoresen@medisin.uio.no)
Anne Johanne Søgaard (Anne.Johanne.Sogaard@fhi.no)

Version: 3 Date: 27 November 2008

Author's response to reviews:

Dear Editor

Thank you for accepting our paper for publication. Before final acceptance you had two editorial requests:

1. This issue was also raised by the reviewers: there are many errors of spelling and grammar in your manuscript. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional copyediting service.

Response: We have used a professional copyediting service to correct the manuscript (OnLine English).

2. The title of your manuscript on our system is different to the one in the copy of your manuscript. Can you please confirm the title of your manuscript, please?

Response: The final title of the manuscript, after the copyediting process is; Three-year follow-up of physical activity in Norwegian youth from two ethnic groups: associations with socio-demographic factors

Yours Sincerely,

Åse Sagatun